Author's response to reviews

Title: A prospective study of the impact of musculoskeletal pain and radiographic osteoarthritis on health related quality of life in community dwelling older people

Authors:

Laura L Laslett (llaslett@utas.edu.au)
Stephen J Quinn (Steve.quinn@flinders.edu.au)
Tania M Winzenberg (Tania.Winzenberg@utas.edu.au)
Kristy Sanderson (Kristy.Sanderson@utas.edu.au)
Flavia Cicuttini (flavia.cicuttini@med.monash.edu.au)
Graeme Jones (Graeme.Jones@utas.edu.au)

Version: 2 Date: 8 December 2011

Author's response to reviews:

I have moved the section on Competing Interests, and Authors Information from where it was at the end of the manuscript to after the end of the Discussion section. I don't have a conclusions subheading as such. We didn't have an acknowledgements subheading as we had no acknowledgements. An acknowledgements subheading has now been added, with the text "We do not have any acknowledgements". You may omit this if you choose.